Author’s response to reviews

Title: More is less, less is more, or does it really matter? The curious case of impact of azacitidine administration schedules on outcomes in patients with myelodysplastic syndromes

Authors:

Rory Shallis (rory.shallis@yale.edu)

Amer Zeidan (amer.zeidan@yale.edu)

Version: 1 Date: 09 Nov 2017

Author’s response to reviews:

The declarations section which includes the mandatory sub-sections listed below has been added as required.

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors’ Contributions
- Acknowledgements
- Authors’ Information

Thank you